Author's response to reviews

Title: Progress towards the achievement of MDG4 in the Commonwealth of Independent States: uncertain data, clear priorities.

Authors:

Adriano Cattaneo (cattaneo@burlo.trieste.it)
Ilkhom Gafurov (gafurov@burlo.trieste.it)
Tamara Bomestar (bomestar@burlo.trieste.it)
Marianna Bacci (mariannabac@hotmail.com)
Sanjiv Kumar (ksanjiv@unicef.org)
Dragoslav Popovic (dpopovic@unicef.org)
Giorgio Tamburlini (tamburlini@burlo.trieste.it)

Version: 3 Date: 4 February 2010

Author's response to reviews:

To the Editor.

Sir,

We have revised our manuscript based on the latest comments we received:
1. The sentence on MMR in the abstract has been better qualified:
2. The discussion on mortality estimates has been improved after discussion with UNICEF officers in the Regional Office for CEE/CIS;
3. A sentence on cross sectoral action has been included in the discussion, as we thought that at this point we could not ignore the effects of the economic crisis. A relevant reference has been added.

We hope that the paper can now be accepted for publication.

Sincerely,

Adriano Cattaneo and Giorgio Tamburlini on behalf of all the authors.